Author’s response to reviews

Title: A data review of airway management in patients with oral cavity or oropharyngeal cancer: A single-institution experiences

Authors:

Gang Zheng (gzheng@mdanderson.org)
Lei Feng (leifeng@mdanderson.org)
Carol M Lewis (cdmlewis@mdanderson.org)

Version: 2 Date: 03 May 2019

Author’s response to reviews:

May 2, 2019

Point to point response to the reviewers’ comments for the manuscript of BANE-D-19-00007R1 “Airway management in patients with oral cavity or oropharyngeal cancer: A single-institution experience”.

BMC Anesthesiology

Dear Editor and Reviewers,

Thank you for the opportunity to revise the manuscript of “Airway management in patients with oral cavity or oropharyngeal cancer: A single-institution experience”. We appreciate the editors/reviewers’ comments and incorporated the edits to improve the quality of the manuscript. The information below provides point-to-point responses to the comments and the corresponding changes in the text are marked with Red Ink. Please let us know if there is any additional information we can provide. Thank you for your time and consideration.

Since the second reviewer did not address concern at this time, all the changes and corrections are made based on Dr. Cummings’s comments (Reviewer 1).
Reviewer reports:

Dr. Kenneth C. Cummings, MD, MS, FASA (Reviewer 1): The authors have thoughtfully responded to my comments. Thank you. My only remaining concern is that the added sections should be reviewed and edited for English grammar.

We thank Dr. Cummings’s comments. All the added sections have been carefully reviewed and edited for their scientific accuracy and the correction of English grammar by all the authors. The revisions/corrections are marked with Red Ink in the paragraphs of:

1. Background (lines 3-4) and Methods (lines 1-2) of the Abstract.

2. The last paragraph in section of “Radiation status and airway management characteristics” (lines 20-22 on page 7), and

3. The last paragraph in Discussion (lines 19-25 on page 10)

Thank you for your time and the consideration for review. I am looking forward to your response.

Kind regards,

Gang Zheng, MD

Associate professor of anesthesiology and perioperative medicine

The University of Texas MD Anderson Cancer Center

1400 Holcombe Boulevard

Faculty Center-Unit 409

Houston, Texas 77030

gzheng@mdanderson.org